Author’s response to reviews

**Title:** Ethical implications of digital communication for the patient-clinician relationship: Analysis of interviews with clinicians and young adults with long term conditions (the LYNC study).

**Authors:**

Agnieszka Ignatowicz (a.ignatowicz@warwick.ac.uk; draignantowicz@gmail.com)

Anne Slowther (A-M.Slowther@warwick.ac.uk)

Patrick Elder (patrickjdelder@gmail.com)

Carol Bryce (c.bryce.1@warwick.ac.uk)

Kathryn Hamilton (kathryn.hamilton@ucl.ac.uk)

Caroline Huxley (carolinehuxley@hotmail.com)

Vera Forjaz (vera.forjaz@nhs.net)

Jackie Sturt (jackie.sturt@kcl.ac.uk)

Frances Griffiths (f.e.griffiths@warwick.ac.uk)

**Version:** 2 **Date:** 12 Feb 2018

**Author’s response to reviews:**

Response to Editors comments

1. Ethics approval and consent to participate For Consent to Participate, please state clearly whether written informed consent was obtained from the participants in this section. If the consent was not in written, please provide a reason why verbal consent to participate was used rather than written consent. Please add this information to the methodology section, and copy the full consent to participate statement into the ‘ethics approval and consent to participate section’.

We have now included a statement that written consent was obtained from all participant in the methodology section and in the research ethics and consent to participate section.

2. Acknowledgements
Please add a section 'Acknowledgements' in Declarations. Please only use this section to acknowledge anyone who contributed towards the article who does not meet the criteria for authorship including anyone who provided professional writing services or materials. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section. If no acknowledgements are necessary then please state 'None'.

We have included an acknowledgements section.

3. Authors contribution

Please ensure that all authors have read and approved the final version and add a statement confirming this in Authors’ Contributions.

Statement added

4. List of abbreviations

As abbreviations are used throughout the paper, we ask that you include a list of abbreviations (before the declarations section) for reference and ease of reading. All abbreviations should still be defined in the text at first use.

We have included a list of abbreviations. We have also spelled out in full three abbreviations previously used in the paper as they were only used once or twice.